Author's response to reviews

Title: Brewers' rice induces apoptosis in azoxymethane-induced colon carcinogenesis in rats via suppression of cell proliferation and the Wnt signaling pathway

Authors:

Tan Bee Ling (tbeeling87@gmail.com)
Norhaizan Mohd. Esa (nhaizan@upm.edu.my)
Heshu Sulaiman Rahman (heshusr77@gmail.com)
Hazilawati Hamzah (hazila@vet.upm.edu.my)
Roselina Karim (rosaz@upm.edu.my)

Version: 4 Date: 5 August 2014

Author's response to reviews: see over
Dear Dr. Tom Rowles,

Submission of Revised Research Article

With reference to the above matter, I am pleased to submit my revised research article to be considered for publication in BMC Complementary and Alternative Medicine. This manuscript presents on the potential of brewers’ rice, a by-product in the rice milling process, in the inhibition of cell proliferation, induce apoptosis, and suppress of COX-2 and β-catenin expression via the Wnt signaling pathway. This manuscript consists of 6 figures and 4 tables. The authors declare that they have no competing interests.

Your kind consideration regarding this matter is highly appreciated. Thank you.

Yours sincerely,

NORHAIZAN MOHD ESA

Corresponding author:
Norhaizan Mohd Esa
Department of Nutrition and Dietetics
Faculty of Medicine and Health Sciences
Universiti Putra Malaysia, MALAYSIA.
Email: nhaizan@upm.edu.my
Phone: +60389472427 Fax: +60389426769
LIST OF CORRECTION

Title: Brewers’ rice induces apoptosis in azoxymethane-induced colon carcinogenesis in rats via suppression of cell proliferation and the Wnt signaling pathway

Editorial comment:

<table>
<thead>
<tr>
<th>No</th>
<th>Comment from the editor</th>
<th>Correction made</th>
<th>Page no.</th>
</tr>
</thead>
<tbody>
<tr>
<td>1.</td>
<td>Please include details in your Methods section on the source of the brewers rice used in your study.</td>
<td>As stated in the paragraph</td>
<td>Page 4, line 93-95</td>
</tr>
</tbody>
</table>